Author's response to reviews

Title: Rapid tests and urine sampling techniques for the diagnosis of urinary tract infection (UTI) in children under five years: a systematic review

Authors:

Penny F Whiting (penny.whiting@bristol.ac.uk)
Marie E Westwood (mew3@york.ac.uk)
Ian S Watt (isw1@york.ac.uk)
Julie Cooper (Julie.cooper@excha.yhs-tr.northu.nhs.uk)
Jos Kleijnen (jk13@york.ac.uk)

Version: 4 Date: 15 March 2005

Author's response to reviews:

The requested formatting changes have been made:

Figures have been cropped and re-submitted
A 'competing interests statement has been added'
Additional files have been appropriately labelled and referenced in the text